Author’s response to reviews

Title: Prospective Blind Comparative clinical study of two point fixation of zygomatic complex fracture using wire and mini plates.

Authors:

Lakshmi N Gandi (lakshmigandi@yahoo.co.in)
Vivekanand S Kattimani (drvivekanandsk@gmail.com)
Sridhar S Meka (mfmsr@ymail.com)
Srinivas P Chakravarthi (drsrinivaschakravarthi@gmail.com)
Amit V Gupta (dramitygupta@gmail.com)

Version: 2 Date: 10 January 2012

Author’s response to reviews:

The formatting of article according to journal format done, with relevant changes necessary for the particular section, like abstract, body text results and conclusion.